Physics 410: Computational Physics: Homework Assignments

This document will be updated throughout the course

Note: All assignments are due at the beginning of class on the specified date.

To ensure that you download the most recent version of homework assignments, it is safest to first clear the disk and memory caches of your browser, or ensure that the Preferences/Advanced/Cache setting of your browser is set so that cached documents are compared to on-line versions every time.

Homework Due Date Topic Problem Set H1 September 18 Unix H2 September 27 Maple H3 October 11 Fortran Programming H4 October 25 Linear Systems H5 November 8 ODEs H6 November 22 Non-linear Equations